Author’s response to reviews

Title: Increased Breast Cancer Risk in Women with Neurofibromatosis Type 1: A Meta-analysis and Systematic Review of the Literature

Authors:

Lorena Suarez-Kelly (lorena.suarez-kelly@osumc.edu)
Lianbo Yu (lianbo.yu@osumc.edu)
David Kline (david.kline@osumc.edu)
Eric Schneider (eschnei1@jhmi.edu)
Doreen Agnese (doreen.agnese@osumc.edu)
William Carson (william.carson@osumc.edu)

Version: 2 Date: 07 Mar 2019

Author’s response to reviews:

Increased Breast Cancer Risk in Women with Neurofibromatosis Type 1: A Meta-analysis and Systematic Review of the Literature

Lorena P. Suarez-Kelly, M.D.; Lianbo Yu, Ph.D.; David Kline, Ph.D.; Eric B. Schneider, Ph.D.; Doreen M. Agnese, M.D.; William E. Carson, MD

Editor’s Comment and Response

1 – “Please correct the grammar in the new sections added to the manuscript. The sentence: Additionally, a study conducted by Frayling et al. evaluating NF1 constitutional mutation types and breast cancer risk in patients with NF1 and breast cancer showed that different NF1 variants demonstrate different risk of breast cancer and that nonsense and missense mutations may be associates with a higher breast cancer risk. [96]
Additionally, a study conducted by Frayling et al. evaluating NF1 constitutional mutation types and breast cancer risk in patients with NF1 and breast cancer showed that different NF1 variants demonstrated different risks of breast cancer and that nonsense and missense mutations may be associated with a higher breast cancer risk. [96]"

Response: Thank you very much for your noticing this error. Correction has been made.